# Lawrence Berkeley National Laboratory

**Recent Work** 

# Title

Of impacts, agents, and functions: An interdisciplinary meta-review of smart home energy management systems research

**Permalink** https://escholarship.org/uc/item/491803j6

**Authors** McIlvennie, C

Sanguinetti, A Pritoni, M

# **Publication Date**

2020-10-01

## DOI

10.1016/j.erss.2020.101555

Peer reviewed

## Of Impacts, Agents, and Functions: An Interdisciplinary Meta-review of Smart Home Energy Management Systems Research

Claire McIlvennie<sup>1,2</sup>, Angela Sanguinetti<sup>3</sup>, Marco Pritoni<sup>4</sup>

<sup>1</sup>Rubenstein School of Environment and Natural Resources, University of Vermont; <sup>2</sup>Gund Institute for Environment, University of Vermont; <sup>3</sup>Consumer Energy Interfaces, University of California, Davis; <sup>4</sup>Lawrence Berkeley National Laboratory

Corresponding author: Claire McIlvennie <u>cmcilven@uvm.edu</u> Rubenstein School of Environment and Natural Resources University of Vermont 81 Carrigan Drive Burlington, VT 05405

#### Abstract

Smart home energy management technologies (SHEMS) have long been viewed as a promising opportunity to manage the way households use energy. Research on this topic has emerged across a variety of disciplines, focusing on different pieces of the SHEMS puzzle without offering a holistic vision of how these technologies and their users will influence home energy use moving forward. This paper presents the results of a systematic, interdisciplinary meta-review of SHEMS literature, assessing the extent to which it discusses the role of various SHEMS components in driving energy benefits. Results reveal a bias towards technical perspectives and controls approaches that seek to drive energy impacts such as load management and energy savings through SHEMS without user or third-party participation. Not only are techno-centric approaches more common, there is also a lack of integration of these approaches with user-centric, information-based solutions for driving energy impacts. These results suggest future work should investigate more holistic solutions for optimal impacts on household energy use. We hope these results will provoke a broader discussion about how to advance research on SHEMS to capitalize on their potential contributions to demand-side management initiatives moving forward.

#### Keywords

Smart home; energy management; control; information; interdisciplinary; energy impacts; demand-side management

#### 1. Introduction

Many new technologies are expected to play a significant role in the ongoing transition towards a clean energy future. Smart home energy management technologies (SHEMS) in particular, such as automated home systems and connected appliances, have long been viewed as promising opportunities to enhance the residential sector's ability to contribute grid resources required to support this transition [1,2] while providing households with valued services like enhanced comfort and convenience [3–5]. As such, visions of the future energy system often conceptualize the residential sector as a powerful distributed

energy resource (DER), enabled by the deployment of SHEMS and coupled with advances in the Internet of Things and broader smart grid capabilities.

With the potential to provide a dynamic combination of production, storage, and flexible demand, many industry stakeholders expect households to become more active participants in the energy system moving forward [6–9]. Yet, exactly how they will participate remains to be seen. As SHEMS continue to emerge uncertainty remains regarding their ability to deliver on hypothesized energy impacts. These uncertainties include what types of energy services SHEMS will be best suited to provide, what functionalities will prove key to unlocking these energy services, and which parties - such as homeowners, utilities, or the technologies themselves - will drive these potential benefits. Seeking to address these uncertainties, a proliferation of research on SHEMS has emerged in the last decade to better understand their role in the grid of the future. However, as noted by Christensen and colleagues [6], "the high degree of 'interpretive flexibility' associated with the 'smart grid' means that is it imbued with very different and sometimes conflicting interpretations of how solutions should be designed" (345). To date, few people have lived in a home defined by smart energy solutions [10] and the research on SHEMS has emerged across a variety of different disciplines. The result has been a scattered body of research that often presents findings on different pieces of the SHEMS puzzle, emphasizing some over others, without offering a clear, holistic vision of how these technologies and their related energy management strategies could impact energy use in the home moving forward.

Seeking to help put the pieces of that puzzle together, this research surveys and analyzes the dominant discourses in SHEMS research on key questions regarding the energy impacts of SHEMS. Specifically, the analysis aimed to assess the disciplinary perspectives involved in this research and the energy impacts, key actors, and technological functionalities expected to play fundamental roles in the future energy system. We believe this research provides several key contributions to the literature. To the authors' knowledge, it is the first review seeking to explicitly assess the extent to which different disciplines contribute to SHEMS research and quantify the relative prevalence of research regarding different pathways through which SHEMS could impact household energy use. This is done by using quantitative metrics to illustrate the frequency at which core SHEMS constructs are discussed in the literature to date. To do so, the authors develop a conceptual model for understanding how SHEMS could deliver energy impacts and use this model to guide an analysis of the current discourse in the field around components of this model. In addition, this research extends the literature through its goals of understanding how different disciplinary perspectives influence the extent to which certain pathways to energy impacts are researched versus others.

We hope the results of this research will provoke a conversation about how to holistically advance research on SHEMS and their contribution to broader energy transitions. We aim to catalyze discussion about gaps in research that need to be addressed to better understand the ability of SHEMS to provide key energy management services and which disciplines might be poised to collectively contribute to those efforts. With these objectives in mind, this paper proceeds as follows: the remainder of section one reviews the background literature on SHEMS, with a focus towards defining the relevant technologies and their key functionalities. Section two presents the author's conceptual model and discusses the method for the review and analysis. Section three presents the results and section four concludes with a discussion regarding paths forward for this field of research.

## 1.1 Background

SHEMS represent a subset of the wider smart home, internet of things, and home energy management product industries [11]. Traditionally, home energy management technologies have been defined as technologies that "enable households to manage their energy consumption by providing information about how they use energy and/or by allowing them (or third parties) to control energy consumption in the home" [12]. In a comprehensive market study of home energy management technologies, Karlin and colleagues [13] identify two key functionalities crucial to the ability for these technologies to augment the way in which households manage energy consumption: the ability to provide control over energy use and information to the user regarding that usage. SHEMS include the portion of this technology space that have the potential to enable both information and control. Technologies typically fall into three high level categories [14]: user interfaces, such as an energy portal or load monitor; smart hardware, like smart plugs or switches, appliances or thermostats; and software platforms that provide home data analytics. As innovation has advanced, electric vehicles, home battery storage, and solar PV are increasingly considered SHEMS as well. Through these functionalities, SHEMS allow for enhanced data flows and services between households and energy service providers [2,15].

Control functionalities refer to the capability to alter energy use through technologically enabled features and intelligence. SHEMS can provide control to homeowners or third parties through an interface (remote control) or algorithmic control strategies (scheduled automation or optimization based on previous consumption data, user preferences, and/or machine learning, i.e. rule-based control) [14]. Common examples include water heaters capable of being remotely controlled by utilities for direct load control programs and smart thermostats with their ability to learn occupant behaviors and adjust set points accordingly. Technologies with these functionalities have long enabled efficient management of energy consumption in the commercial and industrial sectors, thus the assumption their application to the residential sector would also bring about the chance to capture untapped energy savings naturally follows [16].

In addition, SHEMS offer new opportunities to provide the user in-depth information about their energy consumption and better engage them through a deeper understanding of their home. SHEMS can provide information to building occupants through interfaces such as smartphone apps, in-home displays, or displays embedded in smart appliances [17]. Effects of providing energy feedback to consumers with the goal of driving energy savings, primarily through avenues such as bill inserts or home energy reports and feedback-only devices (e.g., in-home monitors without control functionalities), has been well researched [18]. While these strategies have proven relatively successful in realizing some energy savings, they have struggled to deliver long-term, persistent energy savings or behavioral changes [19–21].

New technologies, coupled with advanced metering infrastructure, seem poised to offer the chance to build off the existing work on energy feedback and behavior-based programs generally to provide more in-depth information to users regarding energy systems (consumptions rates, as well as sources, production rates, waste, and direct and indirect personal and societal impacts). These technologies also afford the opportunity to facilitate two-way flows of communication between the user, technology, and third parties such as utilities [22,23]. These new information streams could empower and encourage homeowners to change their behavior to align with the rhythms and needs of the wider energy system

through enhancing the visibility of that system, thereby supporting more informed decisions about energy-consuming behaviors [8,24,25].

Spurred by these functionalities, SHEMS are expected to impact household energy consumption in numerous ways. Many stakeholders emphasize the potential to bring about household energy savings and related cost reductions both to the user and grid at large, although estimates of the magnitude of energy savings range from negative savings to over 25% depending on the product in use and their functionalities [11,15]. Alternatively, and potentially more promising, are the load management capabilities of SHEMS [26,27], referring to their ability to help manage, coordinate, and control the timing of when and how household end-uses consume energy [28]. Indeed, many stakeholders view SHEMS as key to unlocking "flexible demand" in the residential sector capable of matching the variable supply that accompanies increasingly present renewables [29,30].

As these strategies diffuse into the home and increasingly intersect with everyday life, the question of agency has emerged [31]. SHEMS and their enhanced functionalities theoretically open the door to greater participation of numerous agents in the management of residential energy consumption. These agents include utilities and other third parties, residents, or the smart technologies themselves [30]. A spectrum of different visions regarding how to best to deploy these technologies to maximize their energy management potential exist. Research to date has highlighted uncertainty regarding which parties or technologies should be tasked with managing energy use in the home to access the greatest savings potential [29]. As summarized by Christensen, Gram-Hanssen, and Friis [6], "Some argue for remote control with as little active participation from residents as possible...others work with designs that aim to involve residents actively through continuous information about real-time prices" (345).

Diverse disciplinary perspectives underscore these various conceptualizations of how SHEMS will deliver the greatest energy benefits. Within the realm of energy studies, different disciplines have increasingly been shown to influence how researchers view the role of actors within the energy system in driving energy savings. Stern [32] argued that overreliance on one disciplinary perspective (ex. economic models) could lead to inaccurate conceptualizations of energy consumers and result in overlooking promising policy solutions that other disciplines (such as the behavioral sciences) might otherwise shed light on. Recently, work by Moezzi and Luztenhiser [33] and Strengers [34] has more specifically investigated the ways in which various disciplinary perspectives implicate different actors and technologies in residential energy use and lead to numerous hypotheses surrounding why they use energy and solutions to manage that use. While each perspective brings a partial truth to the table [33], such research suggests that viewing issues of home energy use from a multidisciplinary lens could help develop more robust framings of problems, understandings of the agents of change in the system, and resilient solutions moving forward [34].

Yet, to date, "smartness" in grids, technologies, and systems has often been defined in terms of technical potential and advancements that enable things to think and act for people [35]. Research in the energy sector has been continually shown to have a technical skew. Providing a historical perspective, Wilhite and colleagues [36] discuss the early dominance of device-centric approaches to demand-side management beginning in the 1980s that only started to give way to the social sciences when predominantly technical and economic solutions failed to deliver on expected potential. Supporting this

analysis, Sovacool [37] found that social sciences such as psychology, sociology, and public policy made up less than 20% of the research published in energy studies journal articles between 1999 and 2013.

Recent research suggests this trend has continued as innovations in smart technologies have emerged. Darby [27] discusses two dominant narratives in the field, both centered around active technologies and passive users: the first focused on a passive user amid active technologies aimed at providing comfort and convenience, the other centered around home automation for the sake of allowing buildings to provide and receive grid services. Janda and Topuzi [38] describe narratives where smart technologies serve as the "hero" as compared to less common narratives which emphasize learning by society to overcome complex challenges. Those in favor of a more technological approach argue that energy management systems are in a better position than the user to control energy use due to their ability to alleviate uncertainty related to variables such as prices and weather and plan the appropriate response [39] and issues surrounding persistence of behavior change [40].

However, many strategies that include or rely on the user in efforts to positively affect home energy use do exist [41] and automated control strategies will likely not be sufficient to render them obsolete. Innovations in remote sensing and machine learning offer the chance to improve behavior-based demandside management strategies [40], including more effective eco-feedback interfaces that are salient, precise, and motivating [42] and "eco-feedforward" advice and prompts for personalized actions or new routines households could assimilate into their lifestyles [43]. Pilot studies around such programs have already begun to show initial promise [44–46].

As a result, there have been a series of calls for a more holistic, integrated solution to managing energy in the home, developed by embedding technological solutions into a deep understanding of context and users [47,48]. While the smart home has the potential to incorporate different strategies around information and controls to drive energy benefits, a better understanding of which of these strategies (either alone or integrated together) will deliver the greatest energy impacts is needed. Such integrated approaches could find a synergistic relationship between the role of the resident and their technologies [49]. Broader, ongoing changes in the energy system are creating opportunities to radically rethink the approach to demand side management, relationships between users, technologies, and energy providers [3]. As argued by a growing body of authors in the field, taking advantage of this opportunity and not repeating the historical trend of relying on purely technical or automated fixes will be necessary to meet ambitious energy transition goals.

In order to support these efforts, this research aimed to assess the dominant discourses in SHEMS research on key questions regarding their energy impacts. In particular:

- 1. What disciplinary perspectives have contributed to SHEMS research?
- 2. To what degree has the SHEMS literature focused on different types of energy-related impacts?
- 3. To what degree are different agents (e.g., end-user, third party, or the technologies themselves) considered to be driving the energy-related impacts of SHEMS?
- 4. What functionalities underlying the energy impacts of SHEMS have been the focus of research?
- 5. How is disciplinary perspective associated with the types of impacts, agents, and functionalities emphasized in SHEMS research?

We hypothesize that this research will show a skew towards technology-centric, controls approaches to managing energy use despite repeated calls for more interdisciplinary and user-focused energy research over the last several decades. We believe this current focus limits our understanding of SHEMS ability to deliver desired energy impacts, both to the resident and the grid.

## 2. Method

A systematic meta-review of the SHEMS literature was conducted to answer the research questions outlined above. The analysis focused on SHEMS review papers as a proxy for the state of the literature, capable of providing a landscape view of ongoing trends in and discourse dominating the research and researchers in this space. Within the last decade, a multitude of review papers have been written on the topic of SHEMS. Given the expansiveness of the field and the rapid rate at which it is growing, we sought to develop a methodology that would allow us to reasonably assess the whole state of the field spanning all disciplines. The study of review papers allowed this research to capture and assess a broad swath of the literature over a long period of time and fit our goals of understanding the dominant discourses, in terms of who is studying SHEMS and their foci. The next sections describe the search criteria used to identify relevant papers and the method of analysis.

## 2.1 Literature Search

Papers included in the review had to meet three sets of criteria:

- *Review Paper:* For the purpose of this research, "review paper" was defined as a paper dedicated to assessing the literature relating to the implementation of SHEMS. Thus, single study papers were excluded from the sample. If a paper included both a review and the presentation of new research, only the results of the review section were included in the analysis.
- SHEMS and Key Functionalities. Each paper had to include a discussion of SHEMS and at least one of the two key functionalities identified in the literature review above namely information and/or control. Papers that did not discuss at least one of these functionalities were therefore excluded. While there are many strategies available to alter home energy use, this review focused specifically on those that employ SHEMS. For example, reviews that discussed information provided to users through smart phone applications would have been included but reviews that focused on other strategies, such as information provided to users on bills or home audits, were excluded. If a paper discussed both SHEMS and non-SHEMS related strategies it was included in the sample and only those strategies involving SHEMS were considered in the analysis. In addition, the criteria for SHEMS also sought to ensure papers primarily considered energy management in the residential sector. Papers that only considered other sectors (ex. commercial) were included and only the results explicitly related to residential homes were analyzed.
- *Focus on Energy Impacts.* Finally, papers needed a core focus on the ability for these technologies and their functionalities to specifically manage energy consumption once installed in homes and report empirically derived results regarding the impact of SHEMS on home energy use. Since SHEMS are a subset of the broader space of smart home technologies, these criteria

sought the exclude reviews that 1) discuss the broader smart home space without considering the energy savings or management potential of the technologies and 2) solely focus on topics such as security, communications protocols, and device interoperability. While these topics are certainly important for the ability of SHEMS to function optimally in the home, they are tangential to our core focus on understanding the impacts of SHEMS on energy use. In addition, the criteria to discuss empirically derived energy impacts excluded papers that only discussed hypothetical benefits of SHEMS for managing energy. If a review discussed both hypothetical and empirically supported impacts, only the findings for which empirical support was explicitly cited were coded. For the purpose of this analysis, we defined energy impacts as any outcome from the implementation of SHEMS related (directly or indirectly, positively or negatively) to how a given household consumes energy.

Keyword searches were conducted within four scholarly databases: Google Scholar, SCOPUS, Web of Science, and PsychInfo. Keywords included combinations of "home energy management", "smart home", "control", "information", "feedback", and "review". The searches were conducted in September and early October 2018, thus including papers published up until that time. The authors did not restrict the timeframe for review papers, and thus included articles from any year so long as they met the criteria. Journal articles and conference proceedings were included in the final sample, but non-peer reviewed publications, such as white papers and industry reports, were excluded in the final database even if their content otherwise met the criteria. To determine if a paper met the criteria, the abstract, title, and keywords were first reviewed. If necessary, the body of the text was also scanned.

#### **2.2 General Procedure**

Papers identified for inclusion were compiled in a master database for coding. Coding was performed on the "results" of each paper. If a paper had an explicitly titled "results" or "findings" section, that section was coded. However, a large majority of papers in our sample did not have such sections. In these papers, the "results" section was taken as the main body of the text, excluding the introduction, methodology, and conclusion or discussion. The rationale for this decision was supported by reviewing the final paragraph of each introduction section to confirm they outlined or otherwise set up the subsequent sections as a review and the main contribution of the paper. The authors felt the results section provided a consistent unit of analysis across each paper despite their variations in length and structure. All text, tables, and figures in the results sections were included in the analysis and coding.

Each paper in the sample was coded across four primary dimensions corresponding to the research questions, as summarized in **Table 1**. The procedure for the analysis is also visually represented in **Figure 1**. Coding procedures were conducted according to a coding guide, derived deductively from the research questions and hypotheses driving the study and a conceptual model of SHEMS developed by the authors as represented in **Figure 2**. The authors used a quantitative approach to content analysis, recording when each paper met the criteria for a pre-established code level. Similar methodologies have been used in related studies, such as in the work of Sovacool [37]. The following subsections detail a conceptual model that organizes the coding scheme and the coding method.



Figure 1. Flow diagram for search, coding, and analysis.

## 2.3 SHEMS Conceptual Model

**Figure 2** illustrates the authors' conceptual model of SHEMS, which was used to guide the coding scheme. The aim of such a model was to create a holistic visualization of the key components of SHEMS and potential pathways towards realizing energy impacts. The conceptual model ultimately includes three core components: **agents** capable of taking action in the system, **SHEMS functionalities** providing the ability to deliver information and control, and the resulting **energy impacts** driven by the relationships between the agents and SHEMS functionalities.

Within this model, three key agents exist: SHEMS technologies themselves, users, and third parties. Users and SHEMS operate within the home while third parties are primarily external to it. Within the system, any given agent could be either active or passive. Active agents are defined as those components of the system directly driving the creation of the energy impacts, such as a user making behavior changes, a smart appliance optimizing energy use, or a user or third party remotely controlling end uses through SHEMS interfaces. Alternatively, passive agents are relatively unchanging or assumed constant components of the system, such as an in-home display sitting on a counter and providing information, an individual whose behavior is assumed to remain the same, or a third party setting a dynamic pricing scheme and relying on users or SHEMS to react accordingly.

The model further incorporates the two key functionalities as identified by Ford and colleagues [13] - information and control. These functionalities are depicted by the dashed and dotted arrows within the diagram. Dotted arrows represent flows of information between agents and dashed arrows represent control actions.

Finally, the model illustrates potential energy impacts related to SHEMS. The authors hypothesized that five primary energy-related impacts could result from SHEMS: energy savings, load management, cost savings, energy education (i.e., awareness and knowledge), and specific behavior change. These impacts result from interactions between agents and functionalities.

Taken together, **Figure 2**, illustrates different pathways to achieve different energy impacts through the deployment of SHEMS. The colored arrows illustrate the diversity of pathways through which impacts could be generated based on various interactions of agents within the system via information and control functionalities. For example, the orange arrows originate between the SHEMS and user, representing a flow of information between these agents. This information might then lead the user to better understand their energy consumption (orange arrow from the user to the energy education impact) and therefore change their behavior (orange arrow from energy education to behavior change). This behavior change might reduce peak energy use (load management), help conserve energy (energy savings), and each of these impacts could result in a reduction in the user's energy bill (cost savings). As such, the initial information passed between the user and SHEMS could result in all five of the hypothesized impacts of energy education, behavior change, load management, energy saving, and cost savings. The figure illustrates six example pathways but is not an exhaustive account of all possible pathways in order to maintain visual clarity.



Figure 2. Conceptual model representing the relationships between agents, functionalities, and potential energy impacts related to SHEMS.

## 2.4 Coding Guide and Method

The authors developed a coding guide to assess the extent to which each review paper discussed the different components of the conceptual model described above. An overview of the guide is presented in **Table 1** and a more detailed description of the codes are provided in the following sections.

| Research<br>Question   | Procedure  | Coded Constructs   |
|--|--|--|
| <b>Discipline:</b> What<br>disciplinary<br>perspectives have<br>contributed to<br>SHEMS research?                                      | Code the department of<br>each author ( <i>if</i><br><i>department not</i><br><i>available, look to</i><br><i>department of highest</i><br><i>level of education</i> ) and<br>then aggregate all<br>authors to determine<br>code for each paper. | Used the classification scheme developed by Sovacool<br>[37] which includes 20 categories of disciplinary<br>affiliation <sup>1</sup> :<br>Anthropology<br>Business<br>Communication<br>Computer Science<br>Development<br>Economics<br>Energy<br>Engineering<br>Gender<br>Geography<br>Hard Sciences<br>History<br>Law<br>Life Sciences<br>Philosophy<br>Planning/Architecture<br>Political Science<br>Psychology<br>Public Policy<br>Sociology<br>In addition, allowed for "interdisciplinary" affiliation,<br>if 1) author had training in or belonged to two or more<br>departments or 2) if the paper had authors from two or<br>more separate disciplines. |
| <b>Energy impacts:</b><br>To what degree<br>has the SHEMS<br>literature focused<br>on different types<br>of energy-related<br>impacts? | Code types of impacts<br>for which the paper<br>reviews empirical<br>evidence. Each impact<br>coded as "1" if present<br>in the paper, "0"<br>otherwise.   | Energy Savings<br>Example keywords: "energy savings", "conservation",<br>"energy reduction", "energy efficiency"<br>Load Management<br>Example keywords: "load scheduling", "direct load<br>control", "peak time rebates", "load shifting"<br>Cost Savings<br>Example keywords: "reduce energy costs", "cost<br>savings", "reduce total costs"   |

Table 1. Coding guide used to analyze each paper.

<sup>&</sup>lt;sup>1</sup> See Sovacool ([37], pg 4), Table 2 for a more in-depth description of which disciplines were included in each of the 20 high level categories.

|  |  | Energy Awareness and Education<br>Example keywords: "knew their energy use", "using<br>feedback to explore device usage"  |  |  |  |  |
|--|--|---|--|--|--|--|
|  |  | Behavior Changes<br>Example keywords: "energy management behavior",<br>"change their habits", "turn off the air conditioning"   |  |  |  |  |
|  | Code type of studies in<br>which each impact was<br>observed. Each study<br>type coded as "1" if<br>present in the paper, "0"<br>otherwise.  | Field Study<br>Studies on real-world, or "in the wild",<br>implementations of SHEMS, including test homes, pilot<br>studies, and utility interventions<br>Modeling & Simulation |  |  |  |  |
|  |  | Virtual modeling of SHEMS implementations   |  |  |  |  |
| Agents: To what<br>degree are<br>different agents<br>considered to be  | Code types of agents<br>involved in driving<br>energy impacts. Each<br>agent coded as "1" if   | SHEMS<br>Individual appliances (ex. smart thermostat) or entire<br>systems (ex. smart home platform)  |  |  |  |  |
| considered to be<br>driving the energy-<br>related impacts of<br>SHEMS?  | present in the paper, "0" otherwise.   | Users<br>Individuals within the home interacting/adopting the<br>SHEMS  |  |  |  |  |
|  |  | Third Parties<br>Organizations involved in deploying and/or controlling<br>SHEMs in the home, ex. utilities or third party<br>aggregators                                       |  |  |  |  |
|  | Identify whether each<br>agent holds active<br>and/or passive roles in<br>the system.  | Active<br>Actors driving the creation of the energy impacts<br>Passive<br>Relatively static or unchanging actors in a system  |  |  |  |  |
| Functionalities:<br>What<br>functionalities<br>underlying the<br>energy impacts of<br>SHEMS have been<br>the focus of<br>research? | Code type of SHEMS<br>functionalities<br>employed to drive<br>energy impacts. Each<br>high-level functionality<br>(information, control)<br>coded as "1" if present<br>in the paper, "0" | Control-based<br>Rule-based or remote control<br>Information-based<br>Feedback, Prompts   |  |  |  |  |

| otherwise. Further,<br>descriptors of that<br>functionality (rule-<br>based, remote control,<br>feedback, prompts) |  |
|--|--|
| coded as "1" if present<br>in the paper, "0"<br>otherwise.   |  |

To begin, each paper was coded for disciplinary perspective. First, the disciplinary affiliation or training for each of the contributing authors was coded. For this, the authors referred to the methodology used by Sovacool [37] to assess the disciplines contributing to energy studies. To code the discipline of each author, the departmental affiliation stated on the publication was recorded. If no affiliation was stated, then further research was conducted to determine the discipline of the author's highest degree. If neither of these could be identified, the disciplinary code was left blank. These affiliations were then sorted into the twenty disciplinary categories established by Sovacool [37] and indicated in **Table 1**. The designation of interdisciplinary was recorded if an author either listed two or more different departmental affiliations or received a degree in two or more departments. After each author had been coded, those codes were aggregated to develop a single code for each paper. If the discipline of each author on a paper was the same then the paper was assigned that discipline. A designation of interdisciplinary was given if a paper included authors of at least two different disciplines.

Next, the main body of each paper was reviewed to identify the energy impacts resulting from empirical studies of SHEMS. Energy impacts were coded according to the guide presented in **Table 1**. While the authors primarily focused on the impacts included in the conceptual model, additional themes were allowed to emerge through coding "other" relevant themes and reviewed after completion of the coding. For each of the energy impacts identified, the authors then reviewed the text surrounding this impact to assess the methodology of the study being reviewed, agents present in the system, and the functionalities employed in realizing the energy impacts. With regards to study methodology, the authors recorded the type of study reviewed if it was explicitly stated. If it was not stated, this code was left blank. In terms of agents, consistent with the conceptual model, actors were coded as either active or passive and, in theory, a system could have more than one active and/or passive agent. Each impact was reviewed to see if was driven by information and/or control functionalities. If available, the reviewers also noted the type of information and controls employed, as noted in **Table 1**.

For the codes related to the conceptual model, coding only occurred if a given concept was explicitly mentioned in a paper (i.e., not inferred or assumed). Codes were recorded in a binary fashion, i.e. denoted as either present in the article (1) or not (0). In addition, a review article was allowed to be coded for multiple constructs under the same code, i.e. codes were not mutually exclusive. For example, a paper could discuss energy impacts related to both energy savings and behavioral changes, consider examples of SHEMS in which the end-user is both a passive and active agent, or review the results of studies using both modeling and simulation and field study approaches. For the latter example, if a review paper discussed both modeling and field studies, those codes would have each been recorded as "1" (i.e. both

present in the paper). The coding was completed by the primary author, with co-authors reviewing segments to ensure accuracy and consistency.

## 3. Results

An extensive search returned 31 papers that met our criteria, including 22 journal papers and nine conference papers. All papers included in the final database were published between 2009 and 2018, as illustrated in **Figure 3**. Across this timespan, data indicates a relatively consistent stream of publication of SHEMS review papers year over year, with particularly intensive publication periods in 2015 and 2016, with seven papers published in each of those years, respectively. These two years alone account for the publication of nearly 45% of the papers in the sample.



Figure 3. Number of SHEMs review papers published by year.

All 31 papers that met the criteria are listed in **Table 2**, which summarizes the high-level results of the coding analysis. These results, and their interactions, are discussed more fully in each of the sections below.

## 3.1 Disciplinary Perspectives Contributing to SHEMS Research

To begin, each of the identified review papers was coded for disciplinary affiliation. Results of this exercise revealed four high level disciplinary categories in the sample: engineering, computer science, planning and architecture, and interdisciplinary. Together, interdisciplinary and engineering papers constituted the large majority, 81%, of the reviews in the sample, comprising 42% (n = 13) and 39% (n = 12) respectively. The remaining papers came from the planning and architecture (16%, n = 5) and computer science (3%, n = 1) disciplines.

To better understand the perspectives of the interdisciplinary review papers in the sample, the disciplinary affiliations of their individual authors were analyzed. This analysis revealed a diverse array of disciplines contributing to interdisciplinary papers as represented in **Figure 4.** As with the broader sample of papers, engineering affiliations dominate the sample, representing 40% of the contributing authors. Computer scientists also represent a moderately large share of the authorship, representing 23% of the authors contributing to interdisciplinary reviews. Together, these two disciplines make up just over two-thirds, or 63% of the authors. The remaining one third of the authors, stem from a variety of disciplines including interdisciplinary (9%), hard science (8%), psychology (7%), communications (6%), and economics, energy studies, life sciences, and planning and architecture (2% each).

**Table 2.** List of SHEMS reviews identified and the results of the coding analysis. Blank cells indicate a code was not identified in the review.

| Citation | Year      | Discipline             |                   | ENERGY IMPACTS         |                 |                       | AGENTS              |                     |                     | FUNCTIONALITIES |              |              |
|----------|-----------|------------------------|-------------------|------------------------|-----------------|-----------------------|---------------------|---------------------|---------------------|-----------------|--------------|--------------|
|          | Published |                        | Energy<br>Savings | Load<br>Manag<br>ement | Cost<br>Savings | Behavioral<br>Changes | Energy<br>Education | User                | Smart<br>HEMS       | Third<br>Party  | Information  | Control      |
| [50]     | 2009      | Planning/Archite cture | $\checkmark$      |                        |                 | $\checkmark$          | $\checkmark$        | Active              | Passive             |                 | $\checkmark$ |              |
| [51]     | 2009      | Planning/Archite cture | $\checkmark$      |                        | $\checkmark$    |                       | $\checkmark$        | Active              | Passive             |                 | $\checkmark$ |              |
| [52]     | 2010      | Planning/Archite cture | $\checkmark$      |                        |                 | $\checkmark$          | $\checkmark$        | Active              | Passive             |                 | $\checkmark$ |              |
| [53]     | 2012      | Engineering            |                   | $\checkmark$           |                 |                       |                     | Passive             | Active              |                 |              | $\checkmark$ |
| [25]     | 2012      | Planning/Archite cture |                   | $\checkmark$           |                 |                       |                     | Passive             | Active              | Passive         | $\checkmark$ | $\checkmark$ |
| [54]     | 2012      | Engineering            |                   | $\checkmark$           |                 |                       |                     | Passive             | Active              |                 |              | $\checkmark$ |
| [55]     | 2012      | Interdisciplinary      | $\checkmark$      |                        |                 |                       |                     | Passive             | Active              |                 |              | $\checkmark$ |
| [56]     | 2013      | Engineering            | $\checkmark$      | $\checkmark$           | $\checkmark$    | $\checkmark$          | $\checkmark$        | Active &<br>Passive | Active &<br>Passive | Passive         | $\checkmark$ | $\checkmark$ |
| [57]     | 2013      | Interdisciplinary      | $\checkmark$      |                        |                 |                       |                     | Active              | Passive             |                 | $\checkmark$ |              |
| [58]     | 2014      | Computer<br>Science    | $\checkmark$      |                        | $\checkmark$    |                       |                     | Active &<br>Passive | Active &<br>Passive |                 | $\checkmark$ | $\checkmark$ |
| [59]     | 2014      | Interdisciplinary      | $\checkmark$      |                        |                 |                       |                     | Passive             | Active              |                 |              | $\checkmark$ |
| [60]     | 2014      | Engineering            | $\checkmark$      | $\checkmark$           | $\checkmark$    |                       |                     | Active              | Active              | Passive         | $\checkmark$ | $\checkmark$ |
| [39]     | 2015      | Engineering            |                   | $\checkmark$           | $\checkmark$    |                       |                     |                     | Active              | Passive         |              | $\checkmark$ |
| [19]     | 2015      | Interdisciplinary      | $\checkmark$      |                        | $\checkmark$    | $\checkmark$          | $\checkmark$        | Active              | Passive             | Passive         | $\checkmark$ |              |
| [61]     | 2015      | Interdisciplinary      | $\checkmark$      | $\checkmark$           | $\checkmark$    |                       |                     | Passive             | Active              | Passive         |              | $\checkmark$ |
| [62]     | 2015      | Interdisciplinary      | $\checkmark$      | $\checkmark$           | $\checkmark$    |                       |                     | Active              | Active &<br>Passive |                 | $\checkmark$ | $\checkmark$ |
| [63]     | 2015      | Engineering            | $\checkmark$      | $\checkmark$           | $\checkmark$    |                       |                     | Passive             | Active              | Passive         |              | $\checkmark$ |
| [64]     | 2015      | Planning/Archite cture |                   |                        |                 | $\checkmark$          | $\checkmark$        | Active              | Passive             |                 | $\checkmark$ |              |

| [65] | 2016 | Interdisciplinary | $\checkmark$ | $\checkmark$ | $\checkmark$ |              |              | Passive             | Active              | Passive                |              | $\checkmark$ |
|------|------|-------------------|--------------|--------------|--------------|--------------|--------------|---------------------|---------------------|------------------------|--------------|--------------|
| [66] | 2013 | Interdisciplinary | $\checkmark$ | $\checkmark$ | $\checkmark$ |              |              | Active &<br>Passive | Active              | Passive                | $\checkmark$ | $\checkmark$ |
| [67] | 2016 | Engineering       | $\checkmark$ |              |              |              |              | Passive             | Active              | Passive                |              | $\checkmark$ |
| [68] | 2016 | Interdisciplinary |              | $\checkmark$ | $\checkmark$ |              |              |                     | Active              |                        |              | $\checkmark$ |
| [69] | 2016 | Interdisciplinary | $\checkmark$ | $\checkmark$ | $\checkmark$ |              |              | Passive             | Active              | Passive                |              | $\checkmark$ |
| [70] | 2016 | Engineering       | $\checkmark$ | $\checkmark$ |              |              |              | Passive             | Active              |                        |              | $\checkmark$ |
| [71] | 2016 | Engineering       | $\checkmark$ | $\checkmark$ | $\checkmark$ |              |              | Passive             | Active              | Passive                |              | $\checkmark$ |
| [72] | 2016 | Engineering       | $\checkmark$ | $\checkmark$ | $\checkmark$ |              |              | Active &<br>Passive | Active              | Passive                | $\checkmark$ | $\checkmark$ |
| [73] | 2017 | Engineering       | $\checkmark$ | $\checkmark$ | $\checkmark$ |              |              | Active              | Active &<br>Passive |                        | $\checkmark$ |              |
| [74] | 2017 | Interdisciplinary | $\checkmark$ | $\checkmark$ | $\checkmark$ |              |              | Passive             | Active &<br>Passive | Active<br>&<br>Passive |              | $\checkmark$ |
| [75] | 2017 | Interdisciplinary |              | $\checkmark$ |              |              |              | Active              | Passive             | Active                 |              | $\checkmark$ |
| [76] | 2018 | Interdisciplinary |              | $\checkmark$ | $\checkmark$ | $\checkmark$ | $\checkmark$ | Active              | Active &<br>Passive | Passive                |              | $\checkmark$ |
| [77] | 2018 | Engineering       | $\checkmark$ | $\checkmark$ | $\checkmark$ |              |              | Passive             | Active              |                        |              | $\checkmark$ |

**Figure 4** groups these disciplines (excluding the interdisciplinary category) into three high level categories, namely social sciences (shades of blue), technical sciences (shades of red), and traditional sciences (green). Based on these groupings we see that the technical sciences make up the large majority of authors on interdisciplinary reviews (65%) followed by the social sciences (15%), the traditional sciences (12%), and interdisciplinary (9%).



**Figure 4.** Disciplinary breakdown of SHEMS review papers in sample (top pie chart) and disciplinary breakdown of authors contributing to interdisciplinary reviews (bottom pie chart).

Considering the disciplinary contributions over time, the first review paper in the sample was written from the planning and architecture perspective in 2009. From 2009 through 2018, relatively flat or limited growth in the number of review papers published occurs in the computer science and planning and architecture disciplines, as illustrated in **Figure 5** which shows the cumulative number of papers published by discipline over time. With regards to the interdisciplinary papers, the number of reviews published each year initially increases linearly from 2012 when the first review paper was published to 2014, then averages about two to three review papers per year from 2015 onward. The engineering review papers follow a similar trajectory after the first two papers are published in 2012.



Figure 5. Cumulative number of review papers published by year and discipline.

## 3.2 Energy Impacts Discussed in SHEMS Literature

Within the 31 papers reviewed, each of the hypothesized impacts (energy savings, load management, cost savings, behavior change, and energy education and awareness) emerged in at least one paper, as shown in **Figure 6**. No unexpected impacts emerged.



Figure 6. Results related to energy impacts discussed among the sample of review papers.

Results reveal that the most commonly discussed impacts of SHEMS are energy savings, load management, and cost savings, appearing in roughly 74%, 68%, and 61% of the review papers, respectively. With regards to energy savings, a majority of the articles broadly discussed the ability for SHEMS, like connected appliances and in-home displays, to drive energy savings at the household level. Review papers reported a variety of energy savings results, ranging from increases in consumption, no impact on energy use, and energy savings upwards of 60%, depending on whether energy savings were referring to total household or appliance specific consumption. When discussing load management, many review papers commented on the ability for SHEMS to schedule appliance operation or optimize load curves at the level of the household. Here, reviews generally discussed shifting from on- to off-peak hours (and reduction of peak-to-average ratio), with a number of papers seeking to address solutions to rebound peaks, or periods of increased energy use after a demand response event when customers schedule loads at the same time. Although not always explicitly stated, cost reductions ranged from 5% to 74%; although the majority fell in the 20-30% range, dependent on the rate structure, geographic location, and the technologies involved.

The least commonly discussed energy impacts were behavior change (19%) and energy education and awareness (22%). Reported actual or desired behavioral changes included conservation behaviors such as turning off appliances, in addition to general changes in household habits or lifestyles, with one paper noting end-users stated they wanted to change their behavior as a result of information obtained from SHEMS, but felt unable to do so due to lack of control. Somewhat related, the reviews that discussed educational impacts of SHEMS commented on the ability for these technologies to help users better

understand their electricity use patterns or the energy-related consequences of using different end uses. For example, one paper reviewed a study in which a SHEMS device was used as a tool to explore the home and connect actions, such as opening a window, with consequences, like wasting heat.

**3.2.1 Study Methodologies.** As shown in **Figure 7**, the vast majority of energy impacts reported in the review papers were derived from either modeling and simulation experiments or field pilots. For 13% of the papers, the exact study method could not be identified for any of the SHEMS discussed, so this information was not coded. In total, over half of the review papers in the sample (55%) discussed results from modeling and simulation studies, 45% discussed the results of pilot or field studies. 13% of these papers discussed results from both kinds of studies. Looking at the connection between energy impacts and methodologies, certain energy impacts tend to be studied with one type of method over another. For example, behavior change and energy education and awareness were reported based solely on field studies. Alternatively, load management, cost savings, and energy savings tend to skew towards modeling and simulation studies, although are reported based on the results of field studies as well.



Figure 7. Methods used to study the energy impacts of SHEMS.

## 3.3 What Agents Drive Energy Impacts

The analysis also sought to identify what agents are primarily implicated in SHEMS driving energy impacts. Given the existing literature, the authors hypothesized the dominant agents would be SHEMS, users, and third parties. As hypothesized, analysis of the review papers revealed these are the three

primary agents involved in SHEMS research (no unexpected actors emerged from the review). Since discussion of SHEMS technologies was a criteria for inclusion in the research, these agents were identified in 100% of the review papers (n = 31), followed by users in roughly 94% of the papers (n = 29), and third party actors in just under 52% (n = 16).



Figure 8. Type of agents in SHEMS systems and the ways in which they contribute to energy impacts.

**Figure 8** represents the distribution of agents coded across the review papers and the extent to which they were conceptualized as active or passive components of the system. Of the three agents identified, SHEMS was the one most frequently considered an active part of the system, identified as such in 77% of all review papers. Alternatively, these technologies were seen as passive agents in about 42% of the review papers and conceptualized as both active and passive in just under 20% of the sample. The users were the next most common active agent, coded as active in just over 48% of the papers in the sample. Users were conceptualized as passive in just under 55% of the papers and identified at all in 6% of the review papers analyzed. Finally, third parties, primarily referring to utilities or other grid-focused actors, were conceptualized as active the least often. Third parties were not mentioned in roughly 48% of the review papers), typically a utility administering some type of dynamic rate structure. Third parties were only considered an active participant in just under 7% of the reviews, which includes papers where they were coded as both passive and active (3%).



Figure 9. Association between energy impacts and different agents in SHEMS systems.

**Figure 9** shows the correlations between energy impacts and agents as discussed in the review papers. The metric used to quantify correlation was the phi coefficient. This metric is appropriate given the binary nature of the coding data, i.e., a construct was either present (1) or not present (0) in a given review paper. The metric indicates the extent to which specific codes were likely to appear in the same review paper, such that a strong positive correlation suggests both constructs were often discussed in the same papers, while a strong negative correlation suggests that papers tended to discuss one or the other construct, but not both. To orient the reader to the figure, the top left quadrant shows the correlation between different energy impacts. The bottom left quadrant presents the correlation between different agent types. The dark blue cells indicate strong positive correlations and dark red cells indicate strong negative correlations.

From **Figure 9**, a number of observations can be made about the discourse surrounding relationships between energy impacts and the actors in SHEMS. First, considering the relationships between energy

impacts, strong positive correlations emerge between behavior change and energy education and awareness (phi = 0.91), suggesting they are often discussed together. The results also suggest a moderate positive correlation between load management and cost savings (phi = 0.44). On the other hand, load management has a moderate negative correlation to energy education and awareness (phi = -0.45) and weak negative correlation with behavior change (phi = -0.36), suggesting they are not frequently considered in the same papers.

Considering the relationships between different agent types, third parties have a weak correlation with all other agents. This is intuitive since these agents are not well represented in the papers reviewed. Looking at users and SHEMS agents, however, stronger trends are observed. For example, a strong positive correlation emerged between active users and passive SHEMS (phi = 0.75) and a moderate positive relationship between passive users and active SHEMS (phi = 0.6). Conversely, moderately negative correlations exist between active SHEMS and active users (phi = -0.64), and between passive SHEMS and active SHEMS (phi = -0.64), and between passive SHEMS and passive users (phi = -0.54) and between passive users and active users (phi = -0.55).

Finally, considering the interactions between agent types and assessed energy impacts, no strong relationships emerge across these constructs. However, there are numerous moderate relationships. The data suggest moderate negative correlations between active SHEMS and energy education and awareness (phi = -0.63) and active SHEMS and behavior change (phi = -0.52). In addition, moderate positive relationships exist between active SHEMS and load management (phi = 0.62) and active users and both behavior change (phi = 0.51) and energy education and awareness (phi = 0.56). We see roughly the same relationship between passive SHEMS and each of these same two impacts (behavior change, phi = 0.58, and energy education and awareness, phi = 0.64). Results also point to a moderate positive relationship between passive third parties and both load management (phi = 0.39) and cost savings (phi = 0.5). Finally, the results show a moderate negative relationship between passive SHEMS and both load management (phi = -0.39).

#### **3.4 Functionalities Underlying Energy Impacts**

Each paper was additionally coded to understand whether the discourse around impacts stemmed primarily from controls and/or information-based functionalities of SHEMS. As hypothesized, control-based functionalities dominate the discussion of functionalities employed in the review papers, with 81% of the papers in the sample reporting energy impacts derived from controls-based strategies (n = 25). Alternatively, information-based strategies appeared in 48% of the review papers (n = 15). 29% of the reviews discussed both functionalities (n = 9). This breakdown is represented in **Figure 10**.



# TYPES OF SMART HEMS FUCNTIONATLITIES

Figure 10. Types of SHEMS Functionalities identified in the review papers (bottom pie chart) and the types of control strategies employed (top pie chart).

When discussing information-based strategies, the majority of the review papers broadly discussed the effects of feedback or other information from SHEMS without commenting specifically on the strategies employed. When considering the papers discussing controls-based energy impacts, the type of control (either rule or remote) employed was identified when possible. As shown in **Figure 10**, of the papers discussing controls, the majority (76%) commented on rule-based (algorithmic) control while only 24%

discussed remote-control. 20% discussed both forms of control and in 20% of the papers (n = 5), the type of control could not be distinguished.

## 3.5 SHEMS Systems by Disciplinary Perspective

Finally, **Table 3** presents a summary of the constructs related to the conceptual model aggregated by disciplinary affiliation of the paper. Column one designates each of the four disciplines present and the number of articles within each. The resulting columns show what percentage of the total papers in that subsample discussed each of the energy impacts, agents, and functionalities related to SHEMS. Given our sample size we cannot comment on statistically significant results, but in reviewing the evidence several trends begin to emerge. Since only one paper was present from the computer science discipline, results discussed here focus on the remaining three disciplines (interdisciplinary, engineering, and planning & architecture).

With regards to the energy impacts discussed, each discipline shows a similar tendency to discuss energy savings, with a large majority (between 60-70% each) reporting impacts related to energy savings. Consistent with the overall results, these disciplines all show a substantially lower frequency of discussing impacts related to behavior change and energy education. 20% of planning and architecture papers discussed these impacts while only 8% each of interdisciplinary and engineering papers did. The results begin to show more diversity when considering load management and cost savings. Engineering papers were the most likely to report on load management, with over 80% of them doing so, followed by interdisciplinary (62%) and planning and architecture (20%). Alternatively, interdisciplinary papers were the most likely to discuss cost savings (69%), followed by engineering (58%), and planning and architecture (20%).

When discussing the different agents in the system, both interdisciplinary and engineering papers show a significant skew towards discussing active SHEMS and passive users, while this trend is reversed in the planning and architecture subgroup. In particular, nearly all (91%) of engineering papers discussed active SHEMS, followed by 77% of interdisciplinary papers, and only 20% of planning and architecture. Alternatively, only 17% and 38% of engineering and interdisciplinary papers, respectively, discussed SHEMS in a passive light, as compared to 80% of planning and architecture reviews. When discussing users, the reverse trend emerges. Planning and architecture reviews discussed users as active in 80% of reviews and passive in only 20%. On the other hand, interdisciplinary and engineering papers considered systems in which users were active in only 38% and 33% of papers, respectively, and passive in 54% and 67% of reviews, respectively. Consistent with the high level finding of this analysis, third parties were most often considered passive components of the system across the disciplines and discussed as active in only a small minority of interdisciplinary papers.

Similar to the trends observed surrounding agents in the system, interdisciplinary and engineering papers show a distinct skew towards technical, controls-based functionalities of SHEMS. 92% and 85% of engineering and interdisciplinary papers discussed controls functionalities, respectively. On the other hand, only about a third of the papers in each of these disciplines discussed information-based functionalities of SHEMS. The reverse trend emerges in the planning and architecture field, with all papers (100%) discussing information-based functionalities and a minority (20%) discussing controls.

| Discipline                          |                   | F            | Energy Im       | pacts              |               | Agents             | Functionalities  |                 |       |         |
|-------------------------------------|-------------------|--------------|-----------------|--------------------|---------------|--------------------|------------------|-----------------|-------|---------|
|                                     | Energy<br>Savings | Load<br>Mgmt | Cost<br>Savings | Behavior<br>Change | Energy<br>Ed. | SHEMS              | User             | Third<br>Party  | Info. | Control |
| Interdiscipli<br>nary (N=13)        | 69%               | 62%          | 69%             | 8%                 | 8%            | A: 77%<br>P: 38%   | A: 38%<br>P: 54% | A:15%<br>P:46%  | 31%   | 85%     |
| Engineering (N=12)                  | 67%               | 83%          | 58%             | 8%                 | 0%            | A: 91%<br>P: 17%   | A: 33%<br>P: 67% | A: 0%<br>P: 58% | 33%   | 92%     |
| Planning &<br>Architecture<br>(N=5) | 60%               | 20%          | 20%             | 20%                | 20%           | A: 20%<br>P: 80%   | A: 80%<br>P: 20% | A: 0%<br>P: 20% | 100%  | 20%     |
| Computer<br>Science<br>(N=1)        | 100%              | 0%           | 100%            | 0%                 | 0%            | A: 100%<br>P: 100% | A:100%<br>P:100% | A: 0%<br>P: 0%  | 100%  | 100%    |

**Table 3**. Summary of dominant trends in SHEMS discourse by disciplinary perspective. Note, under the "Agents" column, "A" signifies "Active" and "P" signifies "Passive".

## 4. Discussion & Conclusion

This research sought to conduct a systematic, interdisciplinary assessment of the SHEMS literature in terms of the dominant discourses surrounding their potential to impact household energy use. It analyzed 31 review papers written on this topic since 2009, stemming from engineering, computer science, planning and architecture disciplines and interdisciplinary research collaborations. These reviews had a heavy reliance on modeling and simulation methods, especially when estimating potential energy savings and load management. Review papers tended to focus exclusively on either user-centered, informationbased solutions or techno-centric, controls-focused solutions to managing home energy use, with the latter being much more prevalent. Relatedly, energy savings, load management, and cost savings impacts received the most attention, with a minority of papers focusing on energy education and behavior change impacts. As a result of limited integration, few of the numerous potential pathways for SHEMS to impact household energy use, articulated in the conceptual model presented in Figure 2, have received frequent attention in the literature. The following sections reflect on these key findings among others, consider the limitations of the present study, and draw out key considerations for future work. In particular we look at the need for more holistic solutions to home energy management, the role of the social sciences and interdisciplinary research moving forward, and implications of the reliance on modeling and simulation studies in research.

## 4.1 Advancing SHEMS: A Holistic Path Forward

Taken as a whole, the findings of this review point to the need for more holistic approaches to understanding the role of SHEMS in driving energy benefits in the residential sector and tradeoffs between different approaches to doing so. Results show a skew towards technology centric, control-based solutions to managing home energy use and this bias manifests in several ways. First, the analysis indicates a dominant focus on SHEMS as the core and active agent within the system. While users still appear in nearly all papers, they are more frequently considered as passive actors. One unexpected finding from this review was that third party actors were not identified in over half of the review papers and when they were discussed it was almost exclusively in a passive role through the use of dynamic pricing structures. Further, the results indicate that active users and active SHEMS were rarely discussed in the same papers. While this finding may be less surprising in single study papers taking predominantly one or two views of the system, our study reviewed review papers, which typically seek to cover wider swaths of the literature.

The second way this bias manifests is through a reliance on controls approaches to managing energy use as opposed to information-based ones. Although 29% of the review papers did consider both information and controls-based functionalities, these papers mainly discussed them as two separate solutions to managing energy use, not integrated into one program. A further 52% of papers focused on control-based solutions only, with no mention of information-based opportunities surrounding SHEMS. The limited number of papers discussing behavior-based energy impacts logically flows from this finding - if the technology is primarily charged with controlling energy use why does the user need to change?

Revisiting the model presented in Section 2.2, Figure 2 we can see the ways in which this bias conceptually limits the potential SHEMS-related pathways towards energy impacts. Figure 11 presents an illustration of the modified conceptual model excluding third party actors and their associated pathways to energy impacts and Figure 12 shows the same model excluding the user and related pathways. The ways in which this limits the potential opportunities to enact change in the system is immediately apparent. For example, without the user, impacts around energy education and behavior change disappear, which also limits the potential pathways towards achieving load management, energy savings, and cost savings. Without the third parties, the system loses its connection to the external grid and also reduces pathways to energy savings and load management.



Figure 11. Representation of the SHEMS conceptual model excluding third party agents and related pathways to impact.



Figure 12. Representation of the SHEMS conceptual model excluding user agents and related pathways to impact.

While it may turn out to be true that the greatest potential impacts lie in the technical functionalities of SHEMS, there is no way of knowing this is true unless all potential pathways are investigated. History has shown that policy approaches relying on technical potential will only go so far towards delivering energy benefits if the underlying assumptions regarding the role of actors, such as users, are incorrect. Perhaps the best example of this relates to innovations in manual, programmable, and smart thermostats. While programmable thermostats were expected to provide significantly greater energy savings than their manual counterparts, and initially received an ENERGYSTAR label designation, ENERGYSTAR removed its label from the product in 2009 when these savings were not realized [78]. Research has suggested that this gap, between expected and realized savings, was likely related to the fact that estimates of savings potential often underestimated the degree to which individuals already manually manage heating and cooling loads in their home [8]. For example, Malinick and colleagues [78] evaluate how energy savings projections around these programmable thermostats relied on the assumptions that users programmed their thermostats and let them run. Sintov and colleagues [79] recently found users who pre-program their thermostats are just as likely as those who did not to adjust their thermostat on any given day. Such assumptions likely contributed significantly to lower than expected energy savings and ultimately the removal of the ENERYSTAR designation [78]. It is hypothesized that similar assumptions often underlie the reasons research and development pilots, broadly speaking, often do not live up to expectations [38].

The results of our research suggest that, despite such evidence, SHEMS research has continued to make similar assumptions regarding the passive role of customers. This is supported by recent work by Larsen and colleagues [30], who conducted an analysis of the role of smart technologies in district heating systems in Denmark. The study found that smart home technologies appeared to have been developed with the aim of removing uncertainties surrounding unpredictable human behavior through strategies such as automated controls. The authors argued that this ignores both the ability for these technologies to serve as disruptive forces, reconfiguring household dynamics, and for users to override or subvert controls if motivated to do so. Yet, there is no evidence to suggest that new innovations in automated technologies will be exempt from issues surrounding human behavior that previous technologies, like programmable thermostats, have dealt with. Research has consistently shown that users prefer some level of control over their environments and will take actions to regain it where possible [80]. Indeed, some research suggests that energy savings potential around thermostats could be greatest when both users and technologies are involved in the process of creating heating and cooling schedules [46].

In addition to these findings about the user-technology relationship, our findings regarding third parties also feel disengaged from current developments in the industry. In our sample, when third parties were considered, it was primarily to institute dynamic pricing schemes that active SHEMS would then use to optimize or schedule demand accordingly [69,71,74]. Very few articles discuss the role of programs like direct load control or other utility-run demand response programs. These findings seem disconnected from visions of the future that highlight utilities or other third parties as using smart technologies to unlock new opportunities to manage energy use, afford greater penetration of renewables, and provide grid services to support the energy transition more broadly. In the review by Sovacool [37], he notes "…some critiques suggest that a large gap exists between what energy policy researchers think is important, and what business persons, utility commissioners, and policymakers actually think and do."

Our findings here suggest that this could also be the case within the realm of SHEMS research and policy. Yet third party actors such as utilities seem uniquely placed in the energy system in terms of their connections to a wide array of stakeholders, including industry players in emerging technologies, policymakers at the local, regional, and federal levels, and customers across sectors, to help deploy new innovations. Indeed many such actors are currently in the process of conducting field pilots to test out these technologies. Past research from the field of socio-technical transitions has long pointed to the key role of intermediary or middle actors with such connections to help move systems towards more sustainable states [81,82] and this should be further investigated in future SHEMS research.

Taken together, these findings suggest that future exploration of SHEMS should explicitly investigate the potential for multiple active agents to contribute towards the generation of energy benefits and acknowledge that these agents interact within a system, constantly evolving based on interactions, and potentially in unexpected ways. The results of this analysis seem to primarily fall on only a few components of our conceptual model and thus we encourage future work on SHEMS to dive deeper into a more holistic understanding of the relationships represented in **Figures 2**, **11**, **and 12**. To do so will involve critically examining underlying assumptions regarding which components of the system are truly passive and reconsidering how functionalities of SHEMS regarding information and control can holistically be leveraged together.

## 4.2 Role of the Social Sciences and Interdisciplinarity in SHEMS Research

A crucial component of successfully challenging assumptions and taking a more holistic approach to the successful deployment of SHEMS to deliver energy benefits will be the increased participation of more disciplines in SHEMS research. Results of this study reveal a significant underrepresentation of the social sciences in SHEMS work. An extensive search identified no review papers strictly from a social science perspective, and only 15% of authors contributing to interdisciplinary papers had affiliations within the social sciences. This finding is consistent with Sovacool [37] who argued social sciences are vastly underrepresented in contemporary energy research, broadly speaking. In recent years, the literature has seen increasing research contributions in the smart home or SHEMS space by social scientists, including the work of authors such as Strengers and colleagues [8,83,84] and Hargreaves and colleagues [23,85,86]. This research has brought to light important critical perspectives on whether new innovations in the smart home space actually address energy-related challenges in the residential sector [8] and how future research regarding SHEMS functionalities like information can be reframed to engage a wider range of actors and develop new solutions to address energy transitions [23].

While we believe more contributions such as these will be key to the future success of SHEMS, further research in this space should not focus solely on increasing the presence of social sciences alone, but on integrating these perspectives with more technical ones. Surprisingly, the results of the current analysis suggest an impressive amount of interdisciplinary collaboration in this field to date. Roughly 40% of the review papers in the sample came from interdisciplinary collaborations. Compared to the results of Sovacool [37], who finds that less than one in four papers on energy studies reported interdisciplinary affiliations, the findings of this research suggest greater levels of interdisciplinary collaboration occur on the topic of SHEMS relative to energy studies generally. However, even the interdisciplinary perspectives here were significantly skewed towards technical collaborations (e.g., engineering and computer science).

Perhaps as a result, this research revealed largely similar trends across the interdisciplinary and engineering disciplines, with a slightly more pronounced technical, controls-focused skew within the engineering papers.

There is clearly more interdisciplinary work to do on this topic with richer collaborations between disciplines. Managing home energy use will only become more complex as a wider variety of technologies are introduced into the home where they will interact with user behaviors and household practices. Developing robust solutions will require bringing together and integrating a variety of perspectives, in particular greater contribution from the social sciences. Future work should investigate how best to facilitate these cross-disciplinary collaborations, perhaps through the development or expansion of transdisciplinary theories or frameworks such as the conceptual model presented here. An example of such potential frameworks, the development and use of Energy Cultures Framework [69,70] has proven a valuable tool to help communicate across disciplinary boundaries and bring together diverse perspectives on issues of energy use. To our knowledge, this framework has not yet been widely applied to the SHEMS space.

To date, the dominant policy framing around energy efficiency programs in the residential sector has been to focus on or incentivize either technical, widget-based programs such as more efficient products or home audits, or behavior-based approaches [87]. As research in the realm of SHEMS continues to evolve, more unifying, integrated frameworks surrounding home energy management will be needed to bring together all the pieces of the puzzle. In theory, both information-based strategies and technical controls approaches seek to increase the way in which energy use is controlled. Although these strategies often strive to achieve the same goal, they are often researched through different perspectives. Future programs seeking to manage energy use would benefit from one unified framework from which to analyze trade-offs between these two SHEMS functionalities

## 4.3 Role of Modeling & Simulation in Contributing to Our Understanding of SHEMS

Findings of this review confirm existing arguments in the literature, such as those by Darby [27] that much of the information that exists about the potential impacts of smart technologies on home energy management stems primarily from modeling and simulation studies as opposed to real-world pilots and programs. We find a particular skew in how methods are applied to understand different energy impacts. In particular, estimates of the potential of these technologies to spur broader behavior change or educate users on their connection to the energy system are derived solely from field studies in this sample versus the energy impacts of load management, energy and cost savings, which are studied using both modeling and field studies, with a skew towards modeling and simulation.

On one hand, this finding is unsurprising. Such simulations have long supported decision-making in the energy industry, and they hold advantages in being able to generalize findings to multiple building cases, climate zones, and grid networks. It can be expensive and impractical to do many field pilots and, further, many innovations on the grid side are still in development or hypothetical. In these situations, modeling and simulation can serve as a practical way to explore the realms of possibilities with regards to solutions to address grid challenges, yet few robust field pilots exist to support the extent to which modeled impacts hold true "in the wild".

On the other hand, this is problematic in that many modeling and simulation programs don't capture people in particularly realistic ways, as most simulations are defined predominantly through a technical lens with simplistic representations of the human component of the system [88,89]. For example, studies in our review reference users through preferences defined by hot water, thermostat, and other appliance set point boundaries [e.g., 53,55,56,74], with little to no mention of broader psychological or sociological considerations. Alternatively, they frequently consider users as passive agents who do not actively control the appliances managed by the SHEMS, but rather state their preferences and expect these to be met by the technology. This assumption is often unrealistic, and it is a significant departure from how these appliances are controlled today.

While there is growing recognition of this fact, the misrepresentation and simplistic consideration of behavior can lead to inaccurate assessments of energy impacts [90], inherently limiting the capacity for such models to holistically explore promising program opportunities [89,91] or comment on the potentially negative implications of the deployment of SHEMS. The growing field of computational social science (using tools such as agent-based modeling; [92]) affords an exciting chance to enhance these modeling efforts while real-world deployments come online. Such simulation tools both provide the opportunity to create virtual worlds [93] and have been cited as promising opportunities to better understand the dynamics of the emerging smart grid [94,95]. Such models present ways to more realistically explore possible future scenarios for the grid and identify which interventions to fully develop and roll out before investing in costly pilot or field trials. They could prove potentially powerful tools to help develop and test integrated frameworks, such as those called for in Section 4.2. Further, future modeling efforts should seek to work closely with new field pilots as a way to investigate how impacts of smaller field studies could be scaled. Increased partnerships between academia and industry could prove particularly fruitful. Such collaborations would offer insights to draw on experiences on the ground that might not otherwise be published in peer-reviewed venues, provide rich sources of data with which to validate and calibrate models, and work towards addressing the lack of field studies published in the literature.

## 4.4 Limitations

While the results presented here offer some potentially important findings for the SHEMS literature, they should be considered within the context of several limitations to the methodology. The main goal of this review was to study the dominant discourse around the energy impacts of SHEMS rather than the phenomenon itself. As such, the findings do not add to the body of evidentiary work on the energy impacts of SHEMS and underlying mechanisms.

The meta-review method, of reviewing review papers, offers advantages with regard to achieving the study's goal. Specifically, it enabled a survey of the breadth of literature in a growing and expansive field. However, this approach also has several drawbacks. First, summaries of primary studies within a review limit the amount of detail available and therefore prevent a deeper, more nuanced analysis of the details of SHEMS pathways to delivering energy impacts. The meta-review method also relies on the accurate interpretation of primary studies by review authors. Further, given the lag in research process between conducting and publishing primary research and then that research being incorporated into a review article, the perspectives in these review articles is likely missing the most recent research even though the most recent review in this article was from 2018.

In addition, the authors are aware of seminal review papers on the topic of energy feedback from social science perspectives [20,42,96]. These reviews present information crucial to advancing the use of SHEMS via information functionalities, but they did not surface in the literature search for this research, likely because they do not focus on SHEMS (Darby [96] and Karlin and colleagues [20] review studies of energy feedback largely pre-dating SHEMS with control functionalities; Sanguinetti and colleagues [42] focuses on eco-feedback more broadly). This perhaps points to the need for a social science review to take stock of the insights specifically from these perspectives. Although we hope to encourage more interdisciplinary and less siloed work, such a resource would prove a formative reference for the more technical disciplines as they seek to integrate with the work of other disciplines in this field.

Finally, while the analysis here offers an initial glimpse at the disciplines predominantly contributing to SHEMS research, we would like to acknowledge the challenges around assessing disciplinary affiliation. For example, while the departmental affiliation listed on a publication provides a snapshot of an author's perspective, it does not, for example, reveal when scholars have trained across multiple disciplines in their career. Few explicitly interdisciplinary departments or institutes exist at this time, thus it is possible this analysis underestimates the extent to which certain disciplines contribute to SHEMS research.

## 4.5 Conclusion

As the energy system transitions towards a cleaner, more efficient energy future increasingly defined by distributed energy resources, we believe it is imperative that an understanding of the potential contribution of SHEMS is approached from an interdisciplinary, socio-technical perspective. Such an approach will enhance both academic and industry ability to comprehensively assess the pathways towards achieving desired energy benefits and the implications of these innovations for the energy system and the people it supports. The scale of the challenges ahead in transitioning the energy system are too large to impose artificial constraints on the system when seeking solutions. While automation and controls technologies present new, largely unexplored opportunities to rethink the ways in which energy is managed in the home, they are only one piece of the puzzle. As has been argued by many others in this field, we would be remiss not to equally consider the human side of the equation and find ways to holistically bridge the gap. This research provides evidence that, despite calls for greater integration of disciplinary perspectives on this issue, the gap has not yet been bridged. The energy industry stands in a moment of disruption, where the status quo is being challenged. While this brings uncertainty, it also presents an opportunity to reconsider and reframe the ways in which energy policy and related programs are designed and new technologies are deployed.

## 5. Acknowledgements

Claire McIlvennie would like to gratefully acknowledge the financial support provided by the University of Vermont and the National Science Foundation Integrative Graduate Education and Research Traineeship in Smart Grid.

## 6. References

- [1] J. Potter, E. Stuart, P. Cappers, Barriers and Opportunities to Broader Adoption of Integrated Demand Side Management at Electric Utilities: A scoping study, Berkeley, CA, 2018. http://etapublications.lbl.gov/sites/default/files/barriers\_and\_opps\_idsm\_final\_03222108.pdf.
- [2] NEEP, The Smart Energy Home : Driving Residential Building Decarbonization, 2019. https://neep.org/sites/default/files/resources/SmartEnergyHomeDrivingResidentialDecarbonization

Final1.pdf.

- [3] J. Pierce, D.J. Schiano, E. Paulos, Home, habits, and energy: Examining domestic interactions and energy consumption, Conf. Hum. Factors Comput. Syst. - Proc. 3 (2010) 1985–1994. https://doi.org/10.1145/1753326.1753627.
- [4] C. Wilson, T. Hargreaves, R. Hauxwell-Baldwin, Benefits and risks of smart home technologies, Energy Policy. 103 (2017) 72–83. https://doi.org/10.1016/j.enpol.2016.12.047.
- [5] Y. Strengers, L. Nicholls, Convenience and energy consumption in the smart home of the future: Industry visions from Australia and beyond, Energy Res. Soc. Sci. 32 (2017) 86–93. https://doi.org/10.1016/j.erss.2017.02.008.
- [6] T.H. Christensen, K. Gram-Hanssen, F. Friis, Households in the smart grid: existing knowledge and new approaches., in: L. Hansson, U. Holmberg, H. Brembeck (Eds.), Mak. Sense Consum. Sel. from 2nd Nord. Conf. Consum. Res. 2012, Göteborg: Centre for Consumer Science, University of Gothenburg, 2013: pp. 333–348. http://hdl.handle.net/2077/34508.
- [7] M. Goulden, B. Bedwell, S. Rennick-Egglestone, T. Rodden, A. Spence, Smart grids, smart users? The role of the user in demand side management, Energy Res. Soc. Sci. 2 (2014) 21–29. https://doi.org/10.1016/j.erss.2014.04.008.
- [8] S. Tirado Herrero, L. Nicholls, Y. Strengers, Smart home technologies in everyday life: do they address key energy challenges in households?, Curr. Opin. Environ. Sustain. 31 (2018) 65–70. https://doi.org/10.1016/j.cosust.2017.12.001.
- [9] R. Smale, G. Spaargaren, B. van Vliet, Householders co-managing energy systems: space for collaboration?, Build. Res. Inf. 47 (2019) 585–597. https://doi.org/10.1080/09613218.2019.1540548.
- [10] A.G. Paetz, E. Dütschke, W. Fichtner, Smart Homes as a Means to Sustainable Energy Consumption: A Study of Consumer Perceptions, J. Consum. Policy. 35 (2012) 23–41. https://doi.org/10.1007/s10603-011-9177-2.
- [11] A. Sanguinetti, B. Karlin, R. Ford, K. Salmon, K. Dombrovski, What's energy management got to do with it? Exploring the role of energy management in the smart home adoption process, Energy Effic. 11 (2018) 1897–1911. https://doi.org/10.1007/s12053-018-9689-6.
- [12] R. Ford, B. Karlin, A. Sanguinetti, A. Nersesyan, M. Pritoni, Assessing Players, Products, and Perceptions of Home Energy Management, San Francisco, CA, 2016.
- [13] R. Ford, M. Pritoni, A. Sanguinetti, B. Karlin, Categories and functionality of smart home technology for energy management, Build. Environ. 123 (2017) 543–554. https://doi.org/10.1016/j.buildenv.2017.07.020.
- [14] B. Karlin, R. Ford, A. Sanguinetti, J. Gannon, M. Rajukumar, K. Donnelly, Characterization and Potential of Home Energy Management (HEM) Technology, San Francisco, CA, 2015.
- [15] J. King, Energy Impacts of Smart Home Technologies Report A1801, Washington, DC, 2018. https://aceee.org/sites/default/files/publications/researchreports/a1801.pdf.
- [16] R.. Meyers, E.D. Williams, H.S. Matthews, Scoping the potential of monitoring and control technologies to reduce energy use in homes, Energy Build. 42 (2010) 563–569. https://doi.org/10.1016/j.enbuild.2009.10.026.
- [17] B. Karlin, R. Ford, C. Squiers, Energy feedback technology: A review and taxonomy of products and platforms, Energy Effic. 7 (2014) 377–399. https://doi.org/10.1007/s12053-013-9227-5.
- [18] K. Ehrhardt-Martinez, K.A. Donnelly, J.A. "Skip" Laitner, Advanced metering initiatives and residential feedback programs: A meta-review for household electricity-saving opportunities. Report number E105, Washington, D.C, 2010. https://aceee.org/research-report/e105 %0A.
- [19] K. Buchanan, R. Russo, B. Anderson, The question of energy reduction: The problem(s) with feedback, Energy Policy. 77 (2015) 89–96. https://doi.org/10.1016/j.enpol.2014.12.008.
- [20] B. Karlin, J.F. Zinger, R. Ford, The effects of feedback on energy conservation: A meta-analysis, Psychol. Bull. 141 (2015) 1205–1227. https://doi.org/10.1037/a0039650.
- [21] R. Sussman, R. Gifford, W. Abrahamse, Social Mobilization: How to Encourage Action on Climate Change, 2016. https://pics.uvic.ca/sites/default/files/uploads/publications/FINAL Social

mobilization-Sussman Gifford.pdf.

- [22] N. Lange, D. Fredman, M. Phillips, G. Zavaliagkos, G. Choueiter, Flipping the Energy Data Iceberg: Views into the Consumer Benefits and Energy System of the Future through Intelligent and Adaptive High Resolution Metering, in: 2018 ACEEE Summer Study Effic. Build., 2018.
- [23] T. Hargreaves, Beyond energy feedback, Build. Res. Inf. 46 (2018) 332–342. https://doi.org/10.1080/09613218.2017.1356140.
- [24] J. Froehlich, L. Findlater, J. Landay, C. Science, The Design of Eco-Feedback Technology, in: Proc. SIGCHI Conf. Hum. Factors Comput. Syst. (CHI '10), ACM, Atlanta, 2010: pp. 1999–2008. https://doi.org/10.1145/1753326.1753629.
- [25] J. Pierce, E. Paulos, Beyond energy monitors: Interaction, energy, and emerging energy systems, in: Conf. Hum. Factors Comput. Syst. - Proc., 2012: pp. 665–674. https://doi.org/10.1145/2207676.2207771.
- [26] J. Robinson, R. Narayanamurthy, B. Clarin, C. Lee, P. Bansal, National Study of Potential of Smart Thermostats for Energy Efficiency and Demand Response, in: 2016 ACEEE Summer Study Energy Effic. Build., 2016: pp. 1–11.
- [27] S.J. Darby, Smart technology in the home: time for more clarity, Build. Res. Inf. 46 (2018) 140– 147. https://doi.org/10.1080/09613218.2017.1301707.
- [28] P. Alstone, J. Potter, M.A. Piette, P. Schwartz, M.A. Berger, L.N. Dunn, S.J. Smith, M.D. Sohn, A. Aghajanzadeh, S. Stensson, J. Szinai, T. Walter, L. Mckenzie, L. Lavin, B. Schneiderman, A. Mileva, E. Cutter, A. Olson, J. Bode, A. Ciccone, A. Jain, 2025 California Demand Response Potential Study-Charting California's Demand Response Future: Final Report on Phase 2 Results, 2017. https://eta-publications.lbl.gov/sites/default/files/lbnl-2001113.pdf.
- [29] S. Nyborg, I. Røpke, Energy impacts of the smart home conflicting visions, in: Energy Effic. First Found. a Low-Carbon Soc. - Proceeding ECEEE 2011 Summer Study, 2011: pp. 1849–1860.
- [30] S.P.A.K. Larsen, K. Gram-Hanssen, A. Marszal-Pomianowska, Smart home technology enabling flexible heating demand : implications of everyday life and social practices, 2019 ECEEE Summer Study Proc. Is Effic. Sufficient? (2019) 865–874.
- [31] F. Creutzig, J. Roy, W.F. Lamb, I.M.L. Azevedo, W. Bruine De Bruin, H. Dalkmann, O.Y. Edelenbosch, F.W. Geels, A. Grubler, C. Hepburn, E.G. Hertwich, R. Khosla, L. Mattauch, J.C. Minx, A. Ramakrishnan, N.D. Rao, J.K. Steinberger, M. Tavoni, D. Ürge-Vorsatz, E.U. Weber, Towards demand-side solutions for mitigating climate change, Nat. Clim. Chang. 8 (2018) 268–271. https://doi.org/10.1038/s41558-018-0121-1.
- [32] P.C. Stern, Blind Spots in Policy Analysis: What Economics Doesn't Say About Energy Use, Journaly Policy Anal. Manag. 5 (1986) 200–227.
- [33] M. Moezzi, L. Lutzenhiser, What's Missing in Theories of the Residential Energy User, in: 2010 ACEEE Summer Study Energy Effic. Build., 2010: pp. 207–221.
- [34] Y. Strengers, Peak electricity demand and social practice theories: Reframing the role of change agents in the energy sector, Energy Policy. 44 (2012) 226–234. https://doi.org/10.1016/j.enpol.2012.01.046.
- [35] A.S. Taylor, R. Harper, L. Swan, S. Izadi, A. Sellen, M. Perry, Homes that make us smart, Pers. Ubiquitous Comput. 11 (2007) 383–393. https://doi.org/10.1007/s00779-006-0076-5.
- [36] H. Wilhite, E. Shove, L. Lutzenhiser, W. Kempton, The legacy of twenty years of energy demand management: We know more about individual behavior but next to nothing about demand, in: E. Jochem, J. Sathaye, D. Bouille (Eds.), Soc. Behav. Clim. Chang. Mitig., Kluwer Academic Publishers, London, 2000. https://doi.org/10.1007/0-306-48160-x.
- [37] B.K. Sovacool, What are we doing here? Analyzing fifteen years of energy scholarship and proposing a social science research agenda, Energy Res. Soc. Sci. 1 (2014) 1–29. https://doi.org/10.1016/j.erss.2014.02.003.
- [38] K. Janda, M. Topouzi, Telling tales: Using stories to remake energy policy, Build. Res. Inf. 43 (2015) 516–533. https://doi.org/10.1080/09613218.2015.1020217.
- [39] M. Beaudin, H. Zareipour, Home energy management systems: A review of modelling and

complexity, Renew. Sustain. Energy Rev. 45 (2015) 318–335. https://doi.org/10.1016/j.rser.2015.01.046.

- [40] N.D. Sintov, P. Wesley Schultz, Unlocking the potential of smart grid technologies with behavioral science, Front. Psychol. 6 (2015) 1–8. https://doi.org/10.3389/fpsyg.2015.00410.
- [41] P. Grunewald, M. Diakonova, Flexibility, dynamism and diversity in energy supply and demand: A critical review, Energy Res. Soc. Sci. 38 (2018) 58–66. https://doi.org/10.1016/j.erss.2018.01.014.
- [42] A. Sanguinetti, K. Dombrovski, S. Sikand, Information, timing, and display: A design-behavior framework for improving the effectiveness of eco-feedback, Energy Res. Soc. Sci. 39 (2018) 55– 68. https://doi.org/10.1016/j.erss.2017.10.001.
- [43] A. Meinke, M. Hawighorst, A. Wagner, J. Trojan, M. Schweiker, Comfort-related feedforward information: occupants' choice of cooling strategy and perceived comfort, Build. Res. Inf. 45 (2017) 222–238. https://doi.org/10.1080/09613218.2017.1233774.
- [44] J. Bourgeois, J. Van Der Linden, G. Kortuem, B.A. Price, C. Rimmer, Conversations with my Washing Machine: An in-the-wild Study of Demand Shifting with Self-generated Energy, in: Ubicomp, 2014. https://doi.org/10.1145/2632048.2632106.
- [45] D. Pisharoty, R. Yang, M.W. Newman, K. Whitehouse, ThermoCoach: Reducing home energy consumption with personalized thermostat recommendations, in: BuildSys 2015 - Proc. 2nd ACM Int. Conf. Embed. Syst. Energy-Efficient Built, 2015: pp. 201–210. https://doi.org/10.1145/2821650.2821671.
- [46] R. Yang, D. Pisharoty, S. Montazeri, K. Whitehouse, M.W. Newman, How does eco-coaching help to save energy? Assessing a recommendation system for energy-efficient thermostat scheduling, in: UbiComp 2016 - Proc. 2016 ACM Int. Jt. Conf. Pervasive Ubiquitous Comput., 2016: pp. 1176–1187. https://doi.org/10.1145/2971648.2971698.
- [47] R.H. Jensen, Y. Strengers, J. Kjeldskov, L. Nicholls, M.B. Skov, Designing the desirable smart home: A study of household experiences and energy consumption impacts, in: Conf. Hum. Factors Comput. Syst. - Proc., Montreal, 2018: pp. 1–14. https://doi.org/10.1145/3173574.3173578.
- [48] H. Mela, J. Peltomaa, M. Salo, K. Makinen, M. Hilden, Framing smart meter feedback in relation to practice theory., Sustainability. 10 (2018). https://doi.org/10.3390/su10103553.
- [49] R.J. Cole, Z. Brown, Reconciling human and automated intelligence in the provision of occupant comfort, Intell. Build. Int. 1 (2009) 39–55. https://doi.org/10.3763/inbi.2009.0007.
- [50] S.S. Van Dam, C.A. Bakker, J.D.M. Van Hal, The mediating role of home energy management systems, in: First Eur. Energy Effic. Behav. Conf., 2009: pp. 1–20. http://www.eceee.org/conference proceedings/EE and Behaviour/2009/Panel 2/2.202.
- [51] S. Van Dam, Smart and usable home energy management systems, in: Proc. 6th Int. Symp. Environ. Conscious Des. Inverse Manuf., Sapporo, 2009. http://repository.tudelft.nl/assets/uuid:01f0d3d1-d989-4718-b90c-1e731eeb408d/paper EcoDesign conference 2009.pdf.
- [52] S.S. Van Dam, C.A. Bakker, J.D.M. Van Hal, Home energy monitors: Impact over the mediumterm, Build. Res. Inf. 38 (2010) 458–469. https://doi.org/10.1080/09613218.2010.494832.
- [53] A. Kailas, V. Cecchi, A. Mukherjee, A survey of communications and networking technologies for energy management in buildings and home automation, J. Comput. Networks Commun. 2012 (2012). https://doi.org/10.1155/2012/932181.
- [54] A.M. Saklani, H.S.V.S. Kumar Nunna, S. Doolla, Intelligent demand response approach in smart distribution systems: A review, in: 2012 Annu. IEEE India Conf. INDICON 2012, 2012: pp. 1057–1062. https://doi.org/10.1109/INDCON.2012.6420773.
- [55] E. Torunski, R. Othman, M. Orozco, A. El Saddik, A review of smart environments for energy savings, Procedia Comput. Sci. 10 (2012) 205–214. https://doi.org/10.1016/j.procs.2012.06.029.
- [56] D. Geelen, A. Reinders, D. Keyson, Empowering the end-user in smart grids: Recommendations for the design of products and services, Energy Policy. 61 (2013) 151–161. https://doi.org/10.1016/j.enpol.2013.05.107.

- [57] A. Zipperer, P.A. Aloise-Young, S. Suryanarayanan, D. Zimmerle, R. Roche, L. Earle, D. Christensen, P. Bauleo, Electric energy management in the smart home: Perspectives on enabling technologies and consumer behavior., Proc. IEEE. 101 (2013) 2397–2408. https://doi.org/10.1109/JPROC.2013.2270172.
- [58] A.H. Kazmi, M.J. O'grady, D.T. Delaney, A.G. Ruzzelli, G.M.P. O'hare, A review of wirelesssensor-network-enabled building energy management systems, ACM Trans. Sens. Networks. 10 (2014) 43. https://doi.org/10.1145/2532644.
- [59] A. Mahmood, H. Fakhar, S.H. Ahmed, N. Javaid, Home Energy Management in Smart Grid, in: Int. Ind. Inf. Syst. Conf., Chiang Mai, 2013: pp. 1–2. http://arxiv.org/abs/1311.5385.
- [60] K. Patel, A. Khosla, Home energy management systems in future smart grid networks: A systematic review., in: 2015 1st Int. Conf. Next Gener. Comput. Technol. (NGCT-2015)., Dehradun, India, 2015: pp. 479–483. https://doi.org/10.1109/NGCT.2015.7375165.
- [61] I. Hussain, S. Mohsin, A. Basit, Z.A. Khan, U. Qasim, N. Javaid, A review on demand response: Pricing, optimization, and appliance scheduling, Procedia Comput. Sci. 52 (2015) 843–850. https://doi.org/10.1016/j.procs.2015.05.141.
- [62] A.A. Khan, S. Razzaq, A. Khan, K. Fatima, Owais, HEMs and enabled demand response in electricity market: An overview, Renew. Sustain. Energy Rev. 42 (2015) 773–785. https://doi.org/10.1016/j.rser.2014.10.045.
- [63] Z.A. Khan, S. Ahmed, R. Nawaz, A. Mahmood, S. Razzaq, Optimization based individual and cooperative DSM in Smart Grids: A review., in: 2015 Power Gener. Syst. Renew. Energy Technol., Islamabad, 2015: pp. 1–6. https://doi.org/10.1109/PGSRET.2015.7312239.
- [64] G.T. Wilson, T. Bhamra, D. Lilley, The considerations and limitations of feedback as a strategy for behaviour change, Int. J. Sustain. Eng. 8 (2015) 186–195. https://doi.org/10.1080/19397038.2015.1006299.
- [65] H.T. Haider, O.H. See, W. Elmenreich, A review of residential demand response of smart grid, Renew. Sustain. Energy Rev. 59 (2016) 166–178. https://doi.org/10.1016/j.rser.2016.01.016.
- [66] I. Khan, A. Mahmood, N. Javaid, S. Razzaq, D. Khan, M. Ilahi, Home energy management systems in future smart grids., J. Basic Appl. Sci. 3 (2013) 1224–1231.
- [67] D. Lee, C.C. Cheng, Energy savings by energy management systems: A review, Renew. Sustain. Energy Rev. 56 (2016) 760–777. https://doi.org/10.1016/j.rser.2015.11.067.
- [68] G. Lobaccaro, S. Carlucci, E. Lofstrom, A review of systems and technologies for smart homes and smart grids., Energies. 9 (2016). https://doi.org/10.3390/en9050348.
- [69] A. Malik, J. Ravishankar, A review of demand response techniques in smart grids, in: 2016 IEEE Electr. Power Energy Conf. EPEC 2016, Ottowa, 2016: pp. 1–6. https://doi.org/10.1109/EPEC.2016.7771745.
- [70] A. Nanda, C.K. Panigrahi, Review of smart home energy management, Int. J. Ambient Energy. 37 (2016) 541–546. https://doi.org/10.1080/01430750.2015.1004107.
- [71] A. Saad al-sumatic, M.H. Ahmed, M.M.A. Salama, Smart Home Activities: A Literature Review, Electr. Power Components Syst. 42 (2014) 294–305. https://doi.org/10.1080/15325008.2013.832439.
- [72] B. Zhou, W. Li, K.W. Chan, Y. Cao, Y. Kuang, X. Liu, X. Wang, Smart home energy management systems: Concept, configurations, and scheduling strategies, Renew. Sustain. Energy Rev. 61 (2016) 30–40. https://doi.org/10.1016/j.rser.2016.03.047.
- [73] I. Abubakar, S.N. Khalid, M.W. Mustafa, H. Shareef, M. Mustapha, Application of load monitoring in appliances' energy management – A review, Renew. Sustain. Energy Rev. 67 (2017) 235–245. https://doi.org/10.1016/j.rser.2016.09.064.
- [74] B. Celik, R. Roche, S. Suryanarayanan, D. Bouquain, A. Miraoui, Electric energy management in residential areas through coordination of multiple smart homes, Renew. Sustain. Energy Rev. 80 (2017) 260–275. https://doi.org/10.1016/j.rser.2017.05.118.
- [75] N.D. Sintov, W.P. Schultz, Adjustable Green Defaults Can Help Make Smart Homes More Sustainable, SustainabilitySustainability. 9 (2017) 622. https://doi.org/10.3390/su9040622.

- [76] I. Rendroyoko, N.I. Sinisuka, Residential customer demand response program in microgrid system: A survey literature, Int. J. GEOMATE. 14 (2018) 8–14. https://doi.org/10.21660/2018.46.3512.
- [77] H. Shareef, M.S. Ahmed, A. Mohamed, E. Al Hassan, Review on Home Energy Management System Considering Demand Responses, Smart Technologies, and Intelligent Controllers, IEEE Access. 6 (2018) 24498–24509. https://doi.org/10.1109/ACCESS.2018.2831917.
- [78] T. Malinick, N. Wilairat, J. Holmes, L. Perry, Destined to Disappoint: Programmable Thermostat Savings are Only as Good as the Assumptions about Their Operating Characteristics, in: ACEEE Summer Study Energy Effic. Build., 2012: pp. 162–173.
- [79] N.D. Sintov, L. V. White, H. Walpole, Thermostat wars? The roles of gender and thermal comfort negotiations in household energy use behavior, PLoS One. 14 (2019) 8–10. https://doi.org/10.1371/journal.pone.0224198.
- [80] J.K. Day, L. Heschong, Understanding Behavior Potential: the Role of Building Interfaces, in: ACEEE Summer Study Energy Effic. Build., 2016: pp. 1–13.
- [81] P. Kivimaa, Government-affiliated intermediary organisations as actors in system-level transitions, Res. Policy. 43 (2014) 1370–1380. https://doi.org/10.1016/j.respol.2014.02.007.
- [82] Y. Parag, K.B. Janda, More than filler: Middle actors and socio-technical change in the energy system from the "middle-out," Energy Res. Soc. Sci. 3 (2014) 102–112. https://doi.org/10.1016/j.erss.2014.07.011.
- [83] Y. Strengers, Designing eco-feedback systems for everyday life, in: Conf. Hum. Factors Comput. Syst. Proc., Vancourver, 2011: pp. 2135–2144. https://doi.org/10.1145/1978942.1979252.
- [84] Y. Strengers, Smart Energy in Everyday Life: Are You Designing for Resource Man?, ACM Interact. 21 (2014) 24–31. https://doi.org/10.1145/2621931.
- [85] T. Hargreaves, C. Wilson, Smart Homes and Their Users, Human–Comp, Springer, 2017. https://doi.org/10.1007/978-3-319-68018-7\_1.
- [86] T. Hargreaves, C. Wilson, R. Hauxwell-Baldwin, Learning to live in a smart home, Build. Res. Inf. 46 (2018) 127–139. https://doi.org/10.1080/09613218.2017.1286882 To.
- [87] I.M. Hoffman, M.A. Billingsley, S.R. Schiller, C.A. Goldman, E. Stuart, Energy Efficiency Program Typology and Data Metrics: Enabling Multi-State Analyses Through the Use of Common Terminology. LBNL-6370E, Berkeley, 2013. https://emp.lbl.gov/publications/energy-efficiencyprogram-typology.
- [88] E. Delzendeh, S. Wu, A. Lee, Y. Zhou, The impact of occupants' behaviours on building energy analysis: A research review, Renew. Sustain. Energy Rev. 80 (2017) 1061–1071. https://doi.org/10.1016/j.rser.2017.05.264.
- [89] E. McKenna, S. Higginson, P. Grunewald, S.J. Darby, Simulating residential demand response: Improving socio-technical assumptions in activity-based models of energy demand, Energy Effic. 11 (2018) 1583–1597. https://doi.org/10.1007/s12053-017-9525-4.
- [90] C. Andrews, K.P. Lam, Y. Lu, R. Chang, C. Chan, Annex 66:Definition and Simulation of Occupant Behavior in Buildings Definition and Simulation of Occupant Behavior in Buildings, 2014. https://annex66.org/sites/default/files/2018FinalReport/Annex 66 Deliverable - Occupant Behavior Case Study Sourcebook.pdf%0A.
- [91] T. Hong, D. Yan, S. D'Oca, C. fei Chen, Ten questions concerning occupant behavior in buildings: The big picture, Build. Environ. 114 (2017) 518–530. https://doi.org/10.1016/j.buildenv.2016.12.006.
- [92] W. Jager, A. Ernst, Introduction of the special issue "Social simulation in environmental psychology", J. Environ. Psychol. 52 (2017) 114–118. https://doi.org/10.1016/j.jenvp.2017.07.002.
- [93] J. Epstein, R. Axtell, Growing artificial societies: Social science from the bottom up. No Title, Brookings Institution Press MIT Press, Cambridge, 1996.
- [94] J.R. Snape, K.N. Irvine, C. Rynikiewicz, Understanding energy behaviours and transitions through the lens of a smart grid agent based model, in: Energy Effic. First Found. a Low-Carbon Soc. -

Proceeding ECEEE 2011 Summer Study, 2011: pp. 1919–1930. https://dora.dmu.ac.uk/bitstream/handle/2086/7965/8-245 snape reviewed final%5B1%5D.pdf?sequence=1&isAllowed=y.

- [95] C.S.E. Bale, L. Varga, T.J. Foxon, Energy and complexity: New ways forward, Appl. Energy. 138 (2015) 150–159. https://doi.org/10.1016/j.apenergy.2014.10.057.
- [96] S. Darby, The Effectiveness of Feedback on Energy Consumption. A Review for Defra of the Literature on Metering, Billing and Direct Displays, 2006. https://doi.org/10.4236/ojee.2013.21002.